**Author's response to reviews**

**Title:** The ratio of means method as an alternative to mean differences for analyzing continuous outcome variables in meta-analysis: a simulation study.

**Authors:**

Jan O Friedrich (j.friedrich@utoronto.ca)
Neill KJ Adhikari (neill.adhikari@utoronto.ca)
Joseph Beyene (joseph@utstat.toronto.edu)

**Version:** 6  **Date:** 18 May 2008

**Author's response to reviews:**

17 May 2008

Editor

BMC Medical Research Methodology

Dear Sir/Madam:

We have reviewed the text and made some very minor wording corrections. We have also corrected the affiliation of one of the authors and added a grant number to the acknowledgements for another one of the authors.

Sincerely,

Jan O. Friedrich, MD, DPhil
Departments of Medicine and Critical Care
St. Michael’s Hospital and University of Toronto
30 Bond Street, Bond Wing, Room 4-015
Toronto, Ontario
CANADA M5B 1W8
Tel: 416-864-5559
Fax: 416-864-6013
E-mail: j.friedrich@utoronto.ca